Author's response to reviews

Title: Estimating the lifetime economic burden of stroke according to the age of onset in South Korea: a cost of illness study

Authors:

Hye-Young Kang (hykang2@yonsei.ac.kr)
Seung-Ji Lim (sjlim@yuhs.ac)
Hae Sun Suh (haesun.suh@yonsei.ac.kr)
Danny Liew (dliew@medstv.unimelb.edu.au)

Version: 5 Date: 10 August 2011

Author's response to reviews:

We have made the changes to the format of the manuscript as required by the BMC Series Editorial Production Team.